Author's response to reviews

Title: Comparability of localization data in transnasal and transoral esophagogastroduodenoscopy

Authors:

Serhat Aymaz (aymaz@kliniken-koeln.de)
Bernd Krakamp (krakampb@kliniken-koeln.de)
Oliver Kirschberg (oliver.kirschberg@web.de)
Rolf Lefering (rolf.lefering@uni-wh.de)

Version: 4 Date: 10 October 2010

Author's response to reviews:

Conclusions section - done.
Rename in Background- done.
Rename in Methods - done.
Remove all highlightning - done.